Author’s response to reviews

Title: A study on combination of daptomycin with selected antimicrobial agents: In vitro synergistic effect of MIC value of 1 mg/L against MRSA strains

Authors:
Yi-Chien Lee (cimedin7@yahoo.com.tw)
Pao-Yu Chen (chenpaoyu@gmail.com)
Jann-Tay Wang (wangjt1124@ntu.edu.tw)
Shan-Chwen Chang (changsc@ntu.edu.tw)

Version: 3 Date: 17 Apr 2019

Author’s response to reviews:

Dear Academic Editor and Reviewers:

We thank all the reviewers for all the valuable and constructive comments to improve our manuscript entitled "A study on combination of daptomycin with selected antimicrobial agents: In vitro synergistic effect of MIC values of 1 mg/L against MRSA strains”. We have revised the manuscript according to your suggestions. Please see our point-by-point replies below.

Editor Comments

1. Under your Ethics approval and consent to participate heading in the Declarations section, please clarify whether informed consent obtained from participants was written or verbal.

Ans.: Thank you for your comment. We have made some amendments in the Ethics approval and consent to participate section (Page 21, Lines 342-347).

2. We note that the email address for the corresponding author listed on the title page of the manuscript differs from the one mentioned in the editorial submission system. Please clarify which is the correct address for the corresponding author, amending either the title page or our submission data where appropriate.
Ans.: Thank you for your precious comment. We have corrected the email address for the corresponding author (Page 2, Lines 20, 28).

3. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

Ans.: Thank you for your comment. We have submitted our revised manuscript as a single clean copy without any tracked changes or highlighted text.

We appreciate those comments and suggestions to improve the quality of our manuscript. Thank you again for your kind attention to our manuscript and consideration for publishing our manuscript in “BMC Pharmacology and Toxicology”.

Sincerely yours

Jann-Tay Wang, M.D., Ph.D.
Division of Infectious Diseases
Department of Internal Medicine
National Taiwan University Hospital
No.7 Chung-Shan South Road, Taipei 100, Taiwan
Tel.: 886-2-23123456 ext. 65054
Fax.: 886-2-23971412
e-mail: wangjt1124@ntu.edu.tw